Author's response to reviews

Title: Improving mental health of adolescents with Type 1 diabetes: protocol for a randomized controlled trial of the Nothing Ventured Nothing Gained online adolescent and parenting support intervention

Authors:

Naomi J Hackworth (nhackworth@parentingrc.org.au)
Jan Matthews (jmatthews@parentingrc.org.au)
Kylie Burke (k.burke1@uq.edu.au)
Zvezdana Petrovic (zpetrovic@parentingrc.org.au)
Britt Klein (brittklein@iinet.net.au)
Elisabeth A Northam (lis.northam@rch.org.au)
Michael Kyrios (mkyrios@swin.edu.au)
Lisa Chiechomski (lchiechomski@groupwise.swin.edu.au)
Fergus J Cameron (ergus.cameron@rch.org.au)

Version: 2
Date: 20 November 2013

Author's response to reviews:

Please see attached